Author’s response to reviews

Title: Superficial cervical plexus block for management of herpes zoster neuralgia in C3 dermatome: A case report

Authors:

Hye Young Shin (pain2012@daum.net)
Doo Sik Kim (kds0728md@gmail.com)
Sang Su Kim (dhdarkss@naver.com)

Version: 6 Date: 14 June 2013

Author’s response to reviews:

I am sure that our report 'Superficial cervical plexus block for management of herpes zoster neuralgia in C3 dermatome: A case report' is the first case that SCPB is used effectively in C3 HZ neuralgia. Though this case report, we get the advanced knowledge about SCPB so that SCPB may be meaningfully considered and selected as therapeutic option of refractory HZ neuralgia.